Author's response to reviews

Title: The protective effects of ginsenoside Rg1 against hypertension target-organ damage in spontaneously hypertensive rats

Authors:

Hui Chen (chenhui_luo@163.com)
Jun Yin (yinjun826@sina.com)
Yanping Deng (shimbiro@sina.com)
Min Yang (yangmin_7777@163.com)
Lingling Xu (candyxll@163.com)
Fukang Teng (fineskypig1983@163.com)
Defang Li (lidefang@163.com)
Yufan Cheng (chengyufan2009@163.com)
Dong Wang (zhangji-sh@163.com)
Baohong Jiang (jiangbh@mail.shcnc.ac.cn)
Dean Guo (daguo@mail.shcnc.ac.cn)
Tingting Zhang (tingyu3160@163.com)
Xuan Liu (lillianliucn@yahoo.com.cn)
Wanying Wu (wwanying902@163.com)
Shuhong Guan (shuhong_guan@hotmail.com)
Sha Liu (drliusha@163.com)

Version: 4 Date: 20 February 2012

Author's response to reviews: see over
Dear editor,

Thank you for the speedy review of our above-mentioned manuscript (1090611313576360). We appreciate the acceptance of our manuscript and respectfully re-submit a revised version of our manuscript.

In the revised manuscript, we confirmed the order of authors, deleted the running title from the title page, deleted the keywords from page 2, added Email address for every author and checked the whole manuscript according “Authors’s checklist for manuscript formatting”.

Once again, we would express our appreciation for the acceptance of our work again. If any further clarifications or correspondence is needed, please feel free to email me, jiangbh@mail.shcnc.ac.cn.

Respectfully yours,

Baohong Jiang

Shanghai Institute of Materia Medica,
Chinese Academy of Sciences,
Shanghai Zhangjiang Hitech Park,
Haike Road #199, Shanghai 201203, P. R. China
E-mail: jiangbh@mail.shcnc.ac.cn